Author's response to reviews

Title: Study of chemoprevention of asiaticoside on rats and tumor cells

Authors:

Fatma J. Al-Saeedi (Fatimas@hsc.edu.kw)

Version: 4 Date: 27 February 2013

Author's response to reviews: see over
Dear Editor-in-Chief, **BMC Cancer**

I would like to present the following manuscript entitled: "Study of chemoprevention of asiaticoside on rats and tumor cells" for publication. This manuscript has been done by my own work and I am the corresponding author willing to receive your comments and correspondence.

Few studies have been published regarding the subject of this paper. As a result of this study, asiaticoside is effective *in vitro* and *in vivo*.

Currently, the manuscript data and results will not be published, in whole, or in part, in any other journal.

**Note:**

**Potential peer reviewers for my manuscript:**

1- Tripti De, E-mail: triptide@iicb.res.in
2- Yue Dai, E-mail: yuedaicpu@hotmail.com
3- Mayuree H Tantisira, E-mail: Mayuree.t@chula.ac.th
4- Djerrou Z, E-mail: Zouhir21265@yahoo.fr

Sincerely,

**Dr. Fatma Al-Saeedi**

**Contact address:**
Nuclear Medicine Department, Faculty of Medicine.
Kuwait University (Health Sciences Center).
P.O.Box: 24923 postal code/zip 13110 Safat, Kuwait.
Tel: 00965-24986409/97773385.
Fax: 00965-25338936.
Email: Fatimas@hsc.edu.kw